Author’s response to reviews

Title: Association between tumour infiltrating lymphocytes, histotype and clinical outcome in epithelial ovarian cancer

Authors:

Fiona James (jamesfr@hotmail.co.uk)

Mercedes Jiminez-Linan (mercedes.jimenez-linan@addenbrookes.nhs.uk)

Jennifer Alsop (ja503@medschl.cam.ac.uk)

Marie Mack (mbm43@medschl.cam.ac.uk)

Honglin Song (hs310@medschl.cam.ac.uk)

James Brenton (James.Brenton@cruk.cam.ac.uk)

Paul Pharoah (pp10001@medschl.cam.ac.uk)

Raza Ali (Raza.Ali@cruk.cam.ac.uk)

Version: 3 Date: 17 Aug 2017

Author’s response to reviews:

Editor Comments:

1. Thank you for providing a link to https://doi.org/10.17863/CAM.7140. However, we are unable to view the data at this link without requesting permission. We recommend amending the "Availability of data and materials" statement to "The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request". Alternatively, please see our submission guidelines for the statements we can accept in this section at:

   https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials

Apologies, the University data repository service has definitely amended this now and the link now works
2. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Currently, the contributions of authors M.J-L., J.A., H.S. and M.M do not automatically qualify them for authorship. Please provide clarification on their contributions, or remove their names from the list of authors and place them in the “Acknowledgements” section instead.

Acquisition of funding, collection of data, writing of the manuscript or general supervision of the research group, alone, does not usually justify authorship.

We have clarified the contributions of these authors to

"Pathology review: H.R.A. and M.J-L.

Processing of pathology material and generation of pathology data: J.A., H.S. and M.M.

Management of the SEARCH clinical database: H.S."

Many thanks